Author's response to reviews

Title: How does correlation structure differ between real and fabricated data-sets?

Authors:

Dr Noori Akhtar-Danesh (daneshn@mcmaster.ca)  
Mahshid Dehghan-Kooshkghazi (mahshid@ccc.mcmaster.ca)

Version: 5 Date: 26 Sep 2003

Dear Editor,

"How does correlation structure differ between real and fabricated data-sets?"

Thank you very much for accepting the above manuscript for publication in the BMC Medical Research Methodology. The manuscript is now in its final version based on your final comments. The figures are cropped and the details of figures (figure number/ title/ legend) are in a separate page in the manuscript after the References and before the Tables. Once again we would like to thank you and your colleagues at BioMed Central for all of your efforts which made the publication of the manuscript possible.

Sincerely,

Noori Akhtar-Danesh Mahshid Dehghan-Kooshkghazi